

Working Papers RESEARCH DEPARTMENT

California Wildfires, Property Damage, and Mortgage Repayment

Siddhartha Biswas

Federal Reserve Bank of Philadelphia Supervision, Regulation, and Credit Department

Mallick Hossain

Federal Reserve Bank of Philadelphia Supervision, Regulation, and Credit Department

David Zink

Federal Reserve Bank of San Francisco

ISSN: 1962-5361

Disclaimer: This Philadelphia Fed working paper represents preliminary research that is being circulated for discussion purposes. The views expressed in these papers are solely those of the authors and do not necessarily reflect the views of the Federal Reserve Bank of Philadelphia or the Federal Reserve System. Any errors or omissions are the responsibility of the authors. Philadelphia Fed working papers are free to download at: https://philadelphiafed.org/research-and-data/publications/working-papers.

WP 23-05

PUBLISHED March 2023

REVISED November 2023



California Wildfires, Property Damage, and Mortgage Repayment^{*}

Siddhartha Biswas[†] Mallick Hossain[‡] David Zink[§]

November 2023

Abstract

This paper examines the impact of wildfires on mortgage repayment using novel data that combine property-level damages and mortgage performance. We find that 90-day delinquencies were 4 percentage points higher and prepayments were 16 percentage points higher for properties that were damaged by wildfires compared to properties 1 to 2 miles outside of the wildfire perimeter, which suggests higher risks to mortgage markets than found in previous studies. We find no significant changes in delinquency or prepayment for undamaged properties inside a wildfire boundary. Prepayments are not driven by increased sales or refinances, suggesting insurance claims drive prepayment. Almost 40 percent of affected households receive insurance settlements lower than the estimated replacement costs that define coverage limits. This underpayment and the resulting deficits imply that households receive about \$200,000 to \$300,000 less than their entitled amount under California law.

Keywords: wildfires, mortgage, prepayment risk, climate risk, physical risk, underinsurance

JEL Codes: G21; G51; Q54

^{*}We thank Natee Amornsiripanitch, Xudong An, Judson Boomhower, Larry Cordell, Keyoung Lee, Ken Ueda, David Wylie, and participants of the Richmond Fed's Research/Supervision Climate Jamboree, Interagency Risk Quantification Forum, Consumer Finance Institute Hybrid Seminar, Eastern Economic Association Conference, AREUEA Virtual Seminar, OCC Symposium, and WEAI Conference for helpful comments. We also thank Amy Bach, Jay Feinman, and Ken Klein for sharing their expertise on the statutory underpinnings and legal processes associated with homeowners insurance as well as Jill Thomas for sharing the details of the insurance claims process.

Disclaimer: This Philadelphia Fed working paper represents preliminary research that is being circulated for discussion purposes. The views expressed in this paper are solely those of the authors and do not necessarily reflect the views of the Federal Reserve Bank of Philadelphia, Federal Reserve Bank of San Francisco, or the Federal Reserve System. Any errors or omissions are the responsibility of the authors. No statements here should be treated as legal advice. Philadelphia Fed working papers are free to download at https://philadelphiafed.org/research-and-data/publications/working-papers.

[†]Federal Reserve Bank of Philadelphia. Email: siddhartha.biswas@phil.frb.org

[‡]Federal Reserve Bank of Philadelphia. Email: mallick.hossain@phil.frb.org

[§]Federal Reserve Bank of San Francisco. Email: <u>david.zink@sf.frb.org</u>

1 Introduction

From 2017 to 2021, large wildfires in the United States led to \$16.8 billion in damages per year, compared with \$1.2 billion in damages per year during the previous 37 years.¹ Wildfires are particularly concerning in California, as 13 of the state's 20 most destructive fires (by number of structures burned) have occurred since 2017. Negative wealth shocks after these events can delay mortgage payments, restrict borrowers' access to future credit, and increase lenders' exposure to default risk. Insurance and government aid exist to mitigate this default risk, but these protections have been weakening because insurers have been reducing coverage (Dixon et al., 2018; Kaufman, 2021).² Several large insurers in California, totaling 55 percent of the market share in areas with high fire risk, have either stopped writing or capped new policies, citing the rising costs of insuring wildfires (Boomhower et al., 2023). Furthermore, these safeguards may increase prepayments after natural disasters because households can use insurance payouts to pay mortgages early, which may be a suboptimal use of these funds.

This paper studies the types and extent of risk that natural disasters present to borrowers and lenders by evaluating the impact of wildfires on mortgage repayment. Using a novel database of fire damage inspections geographically merged to properties with mortgages, we separately identify the effects of wildfires on properties that are burned and the effects on properties within the fire perimeter that are undamaged. Existing research finds that different perils (e.g., hurricanes, floods, tornadoes) and events (e.g., Hurricanes Harvey and Katrina) lead to small increases in delinquency, default, and foreclosure that are short-lived because of insurance, government aid, and hazard mitigation efforts.³ Our analysis makes three key contributions regarding the economic effects of natural disasters and finds that wildfires present greater risks to mortgage markets than suggested by previous studies.

First, we show that wildfires lead to large increases in the delinquency rates for properties that are damaged. In contrast, the delinquency rates for undamaged properties within the fire perimeter remain statistically unchanged after the fire. These different effects of wildfires on damaged and undamaged properties within a fire perimeter highlight the need to precisely measure treatment groups exposed to a wildfire using property-level damages instead of

¹The median year between 2017 and 2021 had 18 natural disaster events of any peril exceed \$1 billion in damages, leading to \$8.9 billion in damages per event, which is substantially higher than the six "billion-dollar events" leading to \$6.3 billion in damages per event in the median year during the previous 37 years (NOAA, 2022).

 $^{^{2}}$ Cookson et al. (2023) show that crowdfunding and social networks are another source of emergency funding, although these funds tend to accrue to the most well-off.

³See Gallagher and Hartley (2017); Kousky et al. (2020); Ratcliffe et al. (2020); Du and Zhao (2020); Issler et al. (2021); Billings et al. (2022); Panjwani (2022) and An et al. (2022, 2023).

commonly used fire perimeters; our sample shows that 56 percent of properties and 59 percent of mortgages within the perimeter remain undamaged. This measurement error is exacerbated in California, where stricter building standards under stronger wildfire codes have made newer properties significantly less likely to be destroyed from exposure to a wildfire (Baylis and Boomhower, 2021).⁴ Our findings, combined with those of Baylis and Boomhower (2021), suggest an important role for ex-ante disaster mitigation in reducing the physical damages from wildfires and the consequent risks to mortgage markets. To the best of our knowledge, this is the first paper to analyze multiple natural disaster events using property-level damage information.⁵

Second, our paper finds that wildfires substantially increase prepayment, which can be an inefficient use of insurance funds; indicate welfare losses for households; and pose risks to lenders, servicers, and mortgage-backed securities investors. Damages from natural disasters, except for floods and earthquakes, are covered under the home insurance required by mortgage lenders.⁶ Insurance and government aid reduce lenders' exposure to the default risk caused by natural disasters, but households may use assistance to pay off mortgages and other debts. Gallagher and Hartley (2017) and Gallagher et al. (2023) find large reductions in mortgage and credit card balances after a disaster. Prepayment results in lost interest revenue for lenders, which is larger if the disaster occurs during a low prevailing interest rate period. Effectively, insurance shifts some of the default risk associated with natural disasters to prepayment risk, thereby mitigating banks' exposure by a lesser degree than expected.⁷

Third, we provide evidence that the decision to prepay is motivated by insurance settlements that are insufficient to rebuild. After a wildfire, a household may use insurance settlements toward the rebuilding of the damaged home, the payment of the mortgage balance, or the purchase of a new home. The latter two alternatives can lead to a mortgage prepayment. A large increase in prepayment indicates that many affected households do not use their insurance settlements to rebuild the damaged home. Indeed, we show that properties destroyed by fires on average receive settlements that are much lower than the insurer-estimated replacement costs that determine coverage limits. In such cases, households

⁴Fried (2022) develops a model of climate adaptation and similarly finds that adaptation, such as investments in seawalls and stilts, would reduce the damage from more severe and more frequent storms by one-third.

⁵The previously cited papers construct disaster exposure and damage measures at the county, census tract, census block, or zip code levels. Kousky et al. (2020) use property-level flood damages from home inspections, but their analysis is restricted to Hurricane Harvey.

⁶Households may, and in some cases are required to, purchase flood insurance through the National Flood Insurance Program administered by the Federal Emergency Management Agency (FEMA).

⁷Mortgage prepayment is the primary risk for investors holding mortgages securitized by government-sponsored enterprises (GSEs).

face strong incentives to apply insurance funds toward the mortgage balance instead of rebuilding, and the observed increase in prepayment represents a symptom of broader frictions in insurance markets that leave households with large financial losses in the aftermath of a natural disaster.

Specifically, we construct a novel database linking property-level fire damage to mortgage performance and estimate the effect of 79 wildfires in California on mortgage borrowers' likelihood of delinquency (90 days or more past due) and prepayment (full balance repayment ahead of schedule). We combine wildfire perimeters, property-level damage inspection reports for wildfires that burn at least 1,000 acres and damage at least one structure, and mortgage performance data from 24 large banks between 2013 and 2020. Using a difference-in-differences (DiD) design with multiple treatments, we separately identify the impact of wildfires on both damaged and undamaged properties within fire perimeters relative to our control group of properties located 1 to 2 miles outside of the fire perimeters. We estimate an event study specification, which provides evidence in support of the identifying assumption of parallel trends and shows the impact on mortgage repayment in the 24 months after a wildfire.

We find a statistically and economically significant increase in both delinquency and prepayment within the first three months after a fire for damaged properties. Delinquency rates increase by as much as 4 percentage points, and prepayment rates increase by as much as 16 percentage points. Before the fire, the average delinquency rate is 1.35 percent, and the average prepayment rate is 2 percent. We find no significant changes in delinquency or prepayment for undamaged properties located within a fire perimeter. Our reduced-form parameter estimates include the protective effects of insurance and government aid, which suggest an even larger risk if climate scenarios worsen and assistance programs weaken. Particularly in California, greater wildfire risk has led to an increase in policy nonrenewals by insurers and has shifted households toward more expensive Free Access to Insurance Requirements (FAIR) plans that serve as home insurers of last resort and offer basic fire insurance (Dixon et al., 2018; Kaufman, 2021).

Using policy and claims data for fire-related insurance coverage in California, we show that insurance settlements that are too low to cover rebuilding costs after fires provide strong incentives for households to prepay their mortgage with insurance funds instead.⁸ Almost 40 percent of post-fire insurance claims are underpaid, meaning the settlement received is lower than the coverage limit, which in most cases is the insurer's ex-ante estimate of the replacement cost for a destroyed property. Households receive settlements that are 28 percent lower than the anticipated rebuilding costs. A back-of-the-envelope calculation suggests

⁸We rule out other likely deteriminants of prepayment because any increase in sales and refinances for damaged properties is not quick or large enough to result in an immediate increase in prepayment.

that households receive about \$200,000 to \$300,000 less than their entitled amount under California law. Furthermore, we find a larger increase in prepayment for homes that face greater rebuilding costs using a triple-difference design. This result supports survey findings of underinsurance in California, where reconstruction costs exceed existing coverage often because of surges in construction prices after fires (Dixon et al., 2018; United Policyholders, 2022). Public records data confirm this implication, as only 8 percent of owners with a damaged property file a construction permit in the two years following a fire. These findings suggest the increase in prepayment is a symptom of frictions in insurance markets that leave households with settlements that cover their mortgage but not a full rebuild.

Our results show that wildfires pose a greater risk to mortgage markets than the findings of previous research. After large wildfires, McConnell et al. (2021) find no statistical difference in mortgage repayment, while Issler et al. (2021) find small increases in delinquency and foreclosure. We estimate that wildfires lead to a much larger increase in mortgage delinquency. Our results differ because damage inspection data provide a more precise measurement of property-level damages than these studies' use of wildfire burn perimeters. As many properties inside a burn perimeter do not sustain damage, it is likely that measurement error in identifying treatment groups attenuates the impact of wildfires on mortgage repayment estimated by these studies.⁹ Beyond mortgage delinquency, this paper's focus on mortgage prepayment, which identifies the immediate frictions households face when determining whether to rebuild or walk away from a damaged property, complements the other studies' focus on long-run migration and home prices.

In addition to the above contributions to research on consumer finance outcomes after natural disasters, this paper is related to the broader literature on the economic effects of climate risk. Studies of the impact of climate risk on household welfare consider home values, migration, and labor market outcomes. While analyses of the impact of rising sea levels on house prices has yet to reach a consensus, several papers find negative but temporary effects of hurricanes and floods on home prices (Atreya et al., 2013; Ortega and Taṣpinar, 2018; Gibson and Mullins, 2020; Fang et al., 2023; Addoum et al., 2021).¹⁰ Similarly, large hurricanes have a small and transitory negative impact on affected individuals' employment, income, and liquidity (Farrell and Greig, 2018; Deryugina et al., 2018; Groen et al., 2020).

⁹The magnitude of our estimates of the increase in delinquency for undamaged properties within the fire perimeter is similar to the impact on all properties within a fire perimeter estimated by Issler et al. (2021).

¹⁰See Canals-Cerdá and Roman (2021) and Bakkensen et al. (2023) for comprehensive summaries. Several studies find a house price penalty in areas with a high sea-level rise (Bernstein et al., 2019; Baldauf et al., 2020; Keys and Mulder, 2020), while other studies do not (Atreya and Czajkowski, 2019; Murfin and Spiegel, 2020; Hino and Burke, 2021). Heterogeneous climate beliefs may drive differences in the estimated effects of climate risks on home prices and mortgage markets (Bakkensen and Barrage, 2022; Bakkensen et al., 2023).

However, several studies find that large disasters lead to higher out-migration (Deryugina et al., 2018; Boustan et al., 2020; McConnell et al., 2021). From a banking supervision perspective, climate risk may also impact financial systems. Generally, the research on this topic utilizes a stress-testing framework to assess repayment and bank losses under different climate scenarios and primarily focuses on transition risks.¹¹ An et al. (2022) focus on physical risks and find that greater hurricane damages under future climate scenarios lead to a significant increase in mortgage delinquency and a smaller increase in default, primarily mitigated by existing disaster assistance.

In the remainder of this paper, Section 2 describes our data on wildfire-impacted areas and subsequent loan performance. Section 3 outlines our empirical strategy, and Section 4 presents the estimation results. Section 5 discusses insurance underpayment after wildfires. Section 6 highlights the welfare implications of our findings, and Section 7 concludes.

2 Data

2.1 Data Sources

We rely on four data sets. Wildfire burn perimeters are obtained from Monitoring Trends in Burn Severity (MTBS). Parcel-level wildfire damage reports are provided by CAL FIRE's Damage Inspection (DINS) database. Parcel locations are obtained from CoreLogic Solutions (CoreLogic). Finally, mortgage origination and performance history are from FR Y-14M regulatory data. All results presented in the figures and tables of this paper and Appendix are derived from calculations based on these four data sets unless otherwise noted. Details on the contents and usage for each of these data sets are described below.¹²

2.1.1 MTBS Wildfire Perimeters

Monitoring Trends in Burn Severity (MTBS) is an interagency program conducted by the U.S. Geological Survey (USGS) and USDA Forest Service that is dedicated to mapping the spatial extent of wildfires in the United States. MTBS delineates wildfire burn perimeters by utilizing Landsat satellite imagery to analyze changes in burned areas before and after wildfire

¹¹Researchers and central bankers have conducted stress tests that evaluate transition risks in Colombia (Sever and Perez-Archila, 2021), Europe (Battiston et al., 2017), France (Clerc et al., 2021), Mexico (Roncoroni et al., 2021), the Netherlands (Reinders et al., 2020; Vermeulen et al., 2021), and Norway (Grippa and Mann, 2020).

¹²All analysis was done using the high-performance computing cluster hosted by the Center for the Advancement of Data and Research in Economics (CADRE) at the Federal Reserve Bank of Kansas City (Lougee et al., 2018).

events. This data set includes all wildfires that burned over 1,000 acres in the western United States and over 500 acres in the eastern United States from 1984 through 2020. Because we focus only on wildfires for which parcel-level damage reports are also available, our sample of fires is restricted to the subset of California wildfires occurring between 2013 and 2020 that burned at least 1,000 acres and damaged at least one structure. These 79 wildfires are plotted in Figure 1. While fires that burned under 1,000 acres are not included in our sample, the 79 fires included account for 98 percent of the structure damage caused by wildfires in California from 2013 through 2020.

2.1.2 CAL FIRE Damage Inspection Data

CAL FIRE's DINS database identifies the universe of parcels that were damaged or destroyed by wildfires from 2013 through 2020. For each parcel, damage is assessed by in-person visual inspections by teams of damage inspection specialists. The final data set identifies the extent of the damage (no damage, affected, minor, major, destroyed), the assessor parcel number, the parcel location (latitude/longitude), the parcel address, and limited structure characteristics.¹³ Unaffected parcels are not included for most of the fires. Among the parcels that sustained wildfire damage, 92 percent are classified as destroyed (see Figure A.3). Therefore, we do not consider the intensive margin of destruction and instead classify parcels as damaged or undamaged.

2.1.3 CoreLogic Residential Public Records

We use CoreLogic public records to identify parcel-level latitude/longitude locations, addresses, and assessor parcel numbers. For each fire-month, we calculate the distance (in meters) from each parcel to the closest MTBS wildfire perimeter. We retain parcels that fall either within a wildfire perimeter or 1 to 2 miles (~ 1.6 to 3.2 kilometers) outside a wildfire perimeter in a given month. For parcels falling within a wildfire perimeter, we classify them as damaged or undamaged by merging with DINS data. Recall that DINS contains the universe of damaged parcels but generally does not include undamaged parcels. Therefore, we classify any parcel falling within the fire perimeter but not in DINS as undamaged.

Figure 1 displays an example of the parcels included in our analysis and the corresponding damage status for the Camp Fire, which burned over 14,000 single-family residences in Northern California in 2018. As shown, not all properties in the burn perimeter sustain damage, even for the largest fire in our sample. Across all fires, 59 percent of our mortgage sample and 56 percent of our properties sample within burn perimeters do not sustain damage,

 $^{^{13}}$ See Figure A.1 for an example damage report and Figure A.2 for details on each damage category.

underscoring the importance of incorporating DINS data in addition to wildfire perimeters in order to identify damaged parcels.

2.1.4 Federal Reserve FR Y-14M Mortgage Performance

We obtain monthly mortgage performance history and origination characteristics for firstlien loans from FR Y-14M. Large banks submit these data to the Federal Reserve as part of the stress testing program, with our sample including 24 institutions. These data are collected monthly starting in June 2012 and contain monthly loan performance, borrower characteristics, and property characteristics. The monthly performance history allows us to identify loans that are delinquent or have been prepaid. The borrower characteristics include credit score, debt-to-income ratio, loan-to-value (LTV) ratio, loan term, loan purpose, loan amount, and interest rate. The property characteristics include the address, which allows us to identify the property's fire damage treatment status. Table 1 contains borrower-level summary statistics by treatment group in the month before a wildfire.

	Outsic	le Fire	Inside +	Damaged	Inside + U	Jndamaged
	mean	sd	mean	sd	mean	sd
Credit Score	747.42	65.50	748.77	61.67	754.34	58.53
Debt-to-Income	25.77	41.57	23.45	15.92	24.72	18.40
Interest Rate	0.04	0.01	0.04	0.01	0.04	0.01
Loan Term	328.36	72.23	330.52	71.13	333.03	69.57
log(Loan Amount)	12.67	0.69	12.49	0.84	13.00	0.80
LTV	0.64	0.22	0.66	0.24	0.61	0.21
Delinquency Rate	0.01	0.12	0.01	0.10	0.01	0.09
Prepayment Rate	0.02	0.13	0.01	0.11	0.01	0.11
Building Square Footage	$2,\!130.57$	1,049.57	$2,\!158.83$	1,122.48	2,852.04	1,509.40
Year Built	1977.48	19.70	1980.15	17.76	1984.95	15.77
N	69,	831	4,9	983	7,	126

 Table 1: Summary Statistics

Note: Table contains loan-level summary statistics by treatment group in the month before the wildfire (event time t = -1). Credit score, debt-to-income (DTI), loan-to-value (LTV), and loan amount are as of origination. Interest rate varies by month for adjustable rate mortgages. Delinquent is equal to one if the loan is at least three months behind in event time t = -1 and zero otherwise. Prepayment is equal to one if the loan is prepaid in event time t = -1 and zero otherwise.

Figure 1: Donut Design of Regression Framework



Note: Figure illustrates wildfire boundaries for all wildfires between 2013 and 2020. Inset shows the wildfire boundary of the Camp Fire as well as the status of properties matched to that fire. Within the boundary, red dots denote homes that were damaged by the wildfire and green dots denote structures that were not affected by the wildfire. The gray ring is the omitted region of properties less than a mile from the wildfire boundary. The outer ring with blue dots indicates structures that are within 1 to 2 miles of the wildfire boundary.

2.2 Data Construction

We combine the data from Section 2.1 to generate a loan-fire-month data set and a propertyfire-month data set. For the loan-level data set, we match every fire event with all loans on single-family residential properties inside or within 1 to 2 miles of the wildfire boundary. For each loan-fire, we then pull in the loan performance data starting 24 months before the fire through 24 months after the fire. Therefore, for each fire, we have a balanced four-year window of mortgage performance data.¹⁴ For the property-level data set, we additionally include homes for which we do not observe a mortgage in order to track property transactions before and after the fire. We match all single-family residences inside or within 1 to 2 miles of the wildfire boundary to a damage status. As shown in Table 2, our analysis sample includes 81,940 loan-fire observations, which account for 23.5 percent of the 349,252 property-fire observations.¹⁵

Table 2: Sample Composition

Sample	Inside + Damaged	Inside + Undamaged	Outside Fire	Total
Single-Family Houses	23,233	30,124	$295,\!895$	349,252
Mortgages	4,983	7,126	69,833	81,940

Note: The first row of the table decomposes the number of unique property-fire observations we match by combining our property data on single-family houses with fire damage information. "Outside Fire" indicates that a property is within 1 to 2 miles of a fire perimeter. The second row restricts the sample to single-family houses that have a mortgage in the FR Y-14M database and makes up the sample for our mortgage performance analysis.

While this construction is generally clean, there are a few notable complications that must be addressed. First, the same loan may be affected by multiple fires. If these fires are more than 24 months apart, then we treat them almost as separate observations (still including only one loan fixed effect). Our key assumption is that any effects of the wildfire should fade out after two years. Second, loan-fire histories may overlap (e.g., the same loan may be affected by a fire only one year after it was affected by another fire). In these cases, we opt to keep the loans in our analysis, but we drop the pre-fire periods that overlap with the post-fire period of another fire. This does create an unbalanced panel for loans that are affected, but because of the disastrous nature of wildfires, our view is that including post-fire

 $^{^{14}{\}rm Prepayment}$ is considered a terminal state, and we assume that the loan maintains that status for the full duration of our observation window.

¹⁵An estimated two-thirds of California homes have a mortgage, which implies that our mortgage sample accounts for 35 percent of all mortgages in the fire perimeter and surrounding 1- to 2-mile ring (Johnson, 2022).

observations as pre-fire controls for another fire would incorrectly identify those loans as "unaffected" by a wildfire at that point in time.

2.3 Descriptive Analysis

Before proceeding with our empirical strategy, we visually summarize the impact of wildfires on prepayment and delinquency for loans falling within a wildfire perimeter in Figure 2. As shown in the top panel, our sample contains 12,109 parcels inside a wildfire perimeter at the onset of a fire, only 4,983 of which sustain damage. This once again highlights the importance of incorporating parcel-level damage reports into our analysis; it is not reasonable to assume that all parcels falling within a wildfire perimeter are damaged. Finally, the top panel shows the loan status 24 months post-fire.¹⁶ The bottom panel shows that prepayments increase immediately in the month of the fire and further spike one to two months after the fire. Borrowers also become past due on their mortgage within one to two months after the fire, but the overall share of loans past due subsides six months after the fire. Two years after the fire, most of these delinquencies become current or prepay, with minimal loans entering foreclosure.

3 Analysis

Our goal is to examine the impact of wildfire damage to one's house on delinquency and prepayment risk.¹⁷ In order to estimate the causal effect of wildfires on mortgage repayment, we use a difference-in-differences (DiD) framework with varying treatment intensities. We use quasi-experimental variation from California wildfires, which we measure with our unique data on burn perimeters and property-level damage inspections, to disentangle the impact of fires on damaged homes from the effect on undamaged homes that are within the fire perimeters. For both treatment intensities, our control group consists of properties that are 1 to 2 miles outside of the wildfire boundary. We exclude homes in the 1-mile ring outside of the wildfire and utilize a donut design to mitigate bias from spillover effects on homes directly outside of the fire perimeter.¹⁸

 $^{^{16}\}mathrm{Figure}\ \mathrm{A.4}$ provides snapshots of loan performance three and six months after a fire.

¹⁷Throughout this paper, we use "loan" and "home" interchangeably because it is too cumbersome to write "whether loan i was secured by a property that was damaged by a wildfire." We prefer to refer to homes being affected or damaged by wildfires, while loans become delinquent or are prepaid.

¹⁸Issler et al. (2021) discuss that homes in the 1-mile ring outside of the perimeter were often visually exposed to the fire and estimate large externalities of the wildfires on these homes.



Figure 2: Loan Status Flows After Wildfires



Inside + Damaged



Note: Figure illustrates the decomposition of loans within wildfire boundaries based on whether or not the home was damaged by the wildfire. The top panel shows mortgage status two years post-fire (current, delinquent, foreclosed, or prepaid), and the bottom panel tracks monthly loan status.

We use the following framework,

$$Y_{ift} = \beta_1 Damaged_{ift} + \beta_2 Inside Fire Zone_{ift} + \beta_3 X_{ift} + \lambda_{ft} + \lambda_i + \epsilon_{ift},$$

where Y_{ift} is an indicator for our outcome of interest (delinquency or prepayment). $Damaged_{ift}$ is an indicator for whether home *i* was damaged by fire *f* in month *t*. $InsideFireZone_{ift}$ is an indicator for whether home *i* was not damaged but inside the perimeter of wildfire *f* in month *t*. The vector X_{ift} includes time-varying borrower and loan characteristics, such as current interest rate, credit score, and the share of balance remaining.¹⁹ λ_{ft} consists of fire-year-month fixed effects, to control for different time paths for different fires (i.e., separately capture the behavior of nondamaged loans for each fire event). Finally, λ_i is a loan-level fixed effect.

 β_1 and β_2 are our coefficients of interest. β_1 estimates the effect of having one's home damaged by a wildfire on prepayment or delinquency relative to homes 1 to 2 miles outside of the wildfire perimeter. β_2 estimates the effect of having one's home inside a wildfire perimeter (but *not* damaged) on prepayment or delinquency relative to homes 1 to 2 miles outside of the wildfire perimeter. Inside the burn perimeter, even undamaged houses may be affected by the wildfire (e.g., having to evacuate, dealing with damaged infrastructure, job interruptions). On the other hand, homes 1 to 2 miles away from the wildfire boundary should not be affected (or at least have a substantially reduced risk of being affected) by the wildfire.

This identification strategy relies on the parallel trends assumption. To provide some evidence that parallel trends hold in the pretreatment periods, we estimate an event study specification:

$$Y_{ift} = \sum_{t=-24; t\neq-1}^{t=24} \beta_{1t} Damaged_{ift} + \sum_{t=-24; t\neq-1}^{t=24} \beta_{2t} Inside Fire Zone_{ift} + \beta_3 X_{ift} + \lambda_{ft} + \lambda_i + \epsilon_{ift}.$$
(1)

Several recent studies have documented the challenges of using such two-way fixed-effect regressions to identify the average treatment-on-the-treated effect, ATT (Goodman-Bacon, 2021; Callaway and Sant'Anna, 2021; Borusyak et al., 2021; Sun and Abraham, 2021). These papers show that applications often use earlier-treated groups as controls for later-treated groups, which can lead to estimation bias, and may not contain never-treated groups, which can lead to underidentification. To address these concerns, we follow a "stacked regression" design in which λ_{ft} interacts our time fixed effect with each fire event (Cengiz et al., 2019;

¹⁹Our results are similar if we exclude X_{ift} , alleviating concerns that the impact of treatment on these controls may bias estimates of β_1 and β_2 (Caetano and Callaway, 2023; Roth et al., 2023).

Baker et al., 2022; Bradt and Aldy, 2022). Therefore, these threats to identification do not apply to our within-fire comparison because, for each fire, treatment occurs simultaneously at a single date for the treatment group, and the control group remains as the never-treated group.

4 Results

The top panel of Figure 3 shows that there is little evidence of pre-trends in delinquency for loans within the wildfire perimeter (whether damaged or not). We see a dramatic increase in delinquency starting three months after the fire and peaking at 4 percentage points within four to six months after the fire.²⁰ The increase then gradually declines in the six to 24 months after the fire, likely either due to loans curing (from being made whole from insurance payments) or because loans continue into default and are no longer reported as delinquent. For similar homes within the wildfire boundary that are not damaged by the wildfire, there is no significant change in delinquency after the fire. At best, there may be a muted 0.4 percentage point increase in delinquency four months after the fire, but this quickly fades back to be indistinguishable from zero. While there may be some minor effects of wildfire on undamaged houses, those possible effects pale in comparison with the dramatic increase in delinquency seen for damaged houses.

As a benchmark, the 4 percentage point increase in delinquency is almost *three times* the average delinquency rate of 1.35 percent. Such a large increase in delinquency may impact banks' liquidity but nonetheless provides a transitory view of mortgage performance. Therefore, we focus the remainder of the paper on the impact of wildfires on mortgage prepayment, which is the most common terminal state for our sample of loans inside fire perimeters. Within two years after a fire, almost all of the loans inside a fire perimeter are either current or are prepaid (Figure 2). In our sample, only 23 properties are foreclosed and 238 remain delinquent after two years.

The bottom panel of Figure 3 shows that there is little evidence of pre-trends in prepayment for loans within the wildfire perimeter (whether damaged or not). Immediately after the fire, we see a dramatic increase in prepayment of about 16 percentage points; it then gradually declines (but is still elevated) for 24 months after the fire. For similar homes within the wildfire boundary that are not damaged by the wildfire, there is no significant change in prepayment after the fire. As a benchmark, the 16 percentage point increase in delinquency is *eight times* the average prepayment rate of 2 percent. Accumulating these impacts over

 $^{^{20}}$ Since we define delinquency as 90 days past due, three months is the fastest a loan can become delinquent if the fire is the triggering event.



Figure 3: Event Study of Wildfire on Delinquency and Prepayment



Note: Figure illustrates the coefficients for damaged (β_{1t}) and undamaged (β_{2t}) parcels estimated from the event study specification in Equation 1. The top panel presents results for delinquency and the bottom panel presents results for prepayment. Regression controls for fire-month and loan fixed effects. Robust 95%confidence intervals are indicated by the shaded regions.

time, we find that 67 percent of damaged properties prepay within two years after a fire. Properties that are undamaged or outside the fire perimeter prepay at a substantially lower rate of 35 percent and 32 percent, respectively.

4.1 Robustness

To test the robustness of our results, we reestimate our event study specification on two different samples. First, we exclude the Camp Fire in 2018 to ensure that our estimated results are not driven by the largest event. The Camp Fire damaged 2,563 homes in our mortgage sample, representing 40 percent of all mortgages in our sample. Figure A.5 confirms that the pattern of findings is qualitatively similar and the magnitude of the increase in delinquency rates is identical. Even though the magnitude of the increase in prepayment is smaller when we exclude the largest fire in our sample, prepayment rates for homes damaged by all other fires increase by 9 percentage points relative to homes outside of the fire perimeters. Second, Figure A.6 shows that our results are robust to changing our control group to be homes located within 5 to 6 miles (8 to 9.6 kilometers) outside of the fire perimeter. Choosing a control group farther from the fire perimeter reduces the likelihood of spillover effects from the fire on homes outside the perimeter.

4.2 Determinants of Prepayment

In the aftermath of a wildfire, prepayments can potentially be driven by home sales, refinancing, or homeowners insurance payouts. We can infer which of these three channels are driving our prepayment results by observing if the properties associated with prepaid loans in our sample are sold after a fire and the timing of the transactions. Only 34 percent of damaged properties that prepay are sold within two years of a fire. In comparison, 46 percent of undamaged properties and 49 percent of properties outside a fire perimeter that prepay are sold within two years after a fire and refinances after a fire further confirms that the large immediate increase in prepayments is driven by insurance payouts.

We estimate Equation 1 using indicators for whether a property was refinanced or sold in the months leading up to and after a wildfire. Figure 4 shows results from this estimation using two samples: The top panel uses the sample of all properties in the vicinity of a wildfire from CoreLogic public records, and the bottom panel uses the same sample of mortgages from FR Y-14M as our estimation of the impact on mortgage repayment. Results from both samples confirm that sales activity for damaged parcels does not begin to increase until at least four months after a fire. In fact, the sample of properties highlights an immediate decline in sales for damaged properties. Refinance activity among damaged properties does not differ from that of unaffected properties until at least 10 months after a fire, when we see refinancing among damaged properties decline.²¹ An increase in sales occurring more than four months after a fire clearly does not lead to the large spike in prepayment we observe in the first three months after a fire, as shown in Figure 3. Furthermore, the estimated impact of wildfire damage on sales is significantly lower than the 16 percentage point increase in prepayment. However, it is possible that increased sales contribute to the persistently higher prepayment rates for damaged properties after six months, as we see the rate of sales for damaged properties consistently exceeding the rate for properties that are undamaged or outside the fire zone.

5 Insurance

As the change in sales and refinances after fires is insufficient to explain the pronounced increase in prepayments, we turn our attention to households' use of insurance funds to pay off mortgages. A household's decision to pay off a mortgage after a fire implies that the borrower opted to walk away from the damaged property rather than rebuilding it. While households may face a large financial loss from selling a home in a damaged condition, it may still be optimal to sell if proceeds from the sale and insurance claims sufficiently cover the payment of any remaining mortgage balance and the down payment on a new property. Alternatively, rebuilding becomes less attractive if insurance funds do not sufficiently cover rebuild costs or households face delays due to surges in construction demand from large local concentrations of damaged properties. In this section, we examine these mechanisms by describing the role of insurance after wildfires and estimating whether insurance payments are sufficient to rebuild.

5.1 Background

Damages from wildfires are generally covered under homeowners insurance policies.²² Typical policies include coverage for the replacement cost value (RCV) of the dwelling and additional structures of the residence, personal property, living expenses during the loss of use of the residence, personal liability, and medical payments to others. Each of these coverages is limited to a share of the dwelling coverage, known as Coverage A, unless the homeowner

²¹The long-term decline in refinance may be explained by the higher sales rate for these properties. It is less likely that these newly purchased homes would be immediately refinanced. A decline in refinancing further rejects the hypothesis that prepayments are driven by refinancing.

²²Homeowners insurance policies account for 95 percent of all fire-coverage policies on owner-occupied single-family homes in California based on California insurance data detailed in Section 5.2.



Figure 4: Home Sales and Refinancings Around Wildfire Events

(b) Sample of FR Y-14M mortgages affected by wildfire

Note: Figure illustrates the coefficients for damaged (β_{1t}) and undamaged (β_{2t}) parcels estimated from the event study specification in Equation 1, with indicators for refinance and sales, respectively, as the outcome variable. The top panel presents results for the sample of all properties in the vicinity of a fire, and the bottom panel presents results for the sample of mortgages we use to estimate mortgage repayment outcomes. Regression controls for fire-month and loan fixed effects. Robust 95% confidence intervals are indicated by the shaded regions.

opts to purchase additional coverage.²³ Over 88 percent of California homeowners insurance policies hold extended RCV coverage, under which the settlement can exceed the Coverage A limit, commonly by an additional 20, 25, 50, or even 100 percent, to account for any increased cost of construction and compliance with local building codes after a fire. As a condition of holding extended RCV coverage, insurers require that the Coverage A limit equals the RCV (Klein, 2023).

In the aftermath of a fire, the homeowner and insurance adjuster negotiate a settlement. The insurer sends the settlement to the lender to hold in escrow. The lender then releases funds to the borrower, who may use the settlement to rebuild the home, buy or build a home at a different site, or pay the mortgage. If the borrower finds that the insurance funds are not sufficient to rebuild, the borrower must provide contractor quotes to the insurer and request additional funds. However, this process is not simple in practice and faces several frictions.

Under California Insurance Code §2051.5, established into law in 2004 under Assembly Bill 2199, residents are entitled to the lesser of the full extended coverage limit and the quoted costs of rebuilding the home at the original site regardless of the household's decision to rebuild, buy, or newly build a home at a different site.²⁴ Yet, insurance adjusters dispute this right, and as a result, several households do not receive their entitlements (Bach and Wade, 2018). A survey of homeowners after six large California wildfires since 2013 suggests that many homeowners do not receive settlements that are sufficient to rebuild: 42 percent to 66 percent of the respondents are underinsured and would not be fully covered to replace or rebuild their home (United Policyholders, 2022). These figures highlight two mechanisms that leave households without enough funds to rebuild, even if they carry insurance.

First, households may be *underpaid* after a fire, relative to their coverage. In such cases, the negotiated settlement can be below the policy coverage limit, leading to an ex-post deficit. Commonly, insurers initially pay households the Actual Cash Value (ACV), which deducts material depreciation from the estimated replacement costs. Insurers then release the additional funds, up to the lesser of the replacement cost and coverage limit, only if the household rebuilds and provides multiple contractor quotes. This practice runs contrary to California law, which gives residents 24 months to recover their full benefits even if they purchase a home elsewhere, often due to lack of information regarding the law from out of state insurance adjusters and large national insurers. Under the assumption that the Coverage

²³For example, in California, living expenses during the loss of use of the residence are normally limited to 20 percent of the Coverage A limit (California Department of Insurance, 2021). If the loss is related to a state of emergency, insurance will cover at least 24 months of additional living expenses (CA Ins Code §2051.5 (b) (2)).

 $^{^{24}\}mathrm{If}$ the homeowner decides to purchase a new house, the claim entitlement is also capped by the price of the house.

A limit accurately estimates the replacement cost of the structure, claim payments received below the limit would be insufficient to rebuild. However, the settlement may cover the remaining mortgage balance. Therefore, the household decision to prepay may be a symptom of large welfare losses due to lower-than-anticipated claim settlements. Using policy and claims data, we estimate substantial underpayment rates and associated deficits in Section 5.3.

Second, households may be *underinsured*. That is, they carry too little insurance and their Coverage A limits and extended coverages are not enough to cover a full property rebuild. Underinsurance can occur for several reasons. Insurers may be unwilling to write policies in areas with high fire risk, and households may not desire to pay high premiums for the relatively low-probability event of a total loss. While theoretically possible, we think this explanation to be unlikely because 88 percent of policies have some form of extended RCV coverage, therefore requiring the Coverage A limit to be the RCV.²⁵ However, it is possible that the RCV estimate itself may be low or outdated (Dixon et al., 2018). In these cases, the household may think it is fully covered but the quoted replacement cost value would not be enough to replace the destroyed structure, even after accounting for extended coverage (Klein, 2023). Lastly, such a coverage gap is exacerbated by demand surges after fires in areas with a large concentration of property damage, leading to the increased cost of construction materials, transportation, and labor. We examine this last mechanism of demand surges by estimating the heterogeneous impacts of fires in cases with high rebuilding costs in Section 5.4.

5.2 Policy and Claims Data

We gather data on fire policies and claims in California from 2018 to 2021 from the California Department of Insurance (CDI).²⁶ The data include information on all policies in force in a given year for insurers that write \$10 million or more in premiums for property or homeowners insurance. The data are repeated cross sections of policies that can be linked to the claims made on each policy and cover over 98 percent of the voluntary market. The policy data include coverage type, limits, premiums, deductibles, and start and end dates, and the claims

²⁵In addition, for mortgages secured by government-sponsored enterprises (GSEs), the coverage limit on homeowners insurance needs to cover the lesser of the RCV and the unpaid principal balance of the mortgage. If the mortgage balance is lower than the RCV, the insurance must cover the greater of mortgage balance and 80 percent of RCV.

²⁶The data are collected through Public Records Request GOVR-2023-00074. Every other year, the CDI collects policy and loss data from insurers for the previous two years through the Personal Property Experience (PPE) data call. Our data request includes two such data calls that cover information from 2018 to 2021.

data include the type, date, and amount of loss. Notably, the date reveals the timing of the loss but not the timing of the claim filing or payment.

We focus our sample on fire coverage policies for single units through homeowner or standalone dwelling-fire insurance.²⁷ For each policy in a specific month, we merge in the total dollars of claims associated with a loss incurred in that month to account for settlements received over multiple payments. Lastly, we only include fire-related claims on the primary structures (type A) that are classified as a catastrophic loss.²⁸

The primary limitation of these data is the lack of geographic information. While we observe each policy's zip code, we are not able to link the policy or claim to a specific property in the damage inspection data from CAL FIRE. Therefore, we do not observe the coverage or claims for a specific damaged property. Instead, we identify the set of claims made on the set of damaged properties in a zip code in any particular month.

First, we rank all claims from highest to lowest within zip code j and month t. Then, we keep the top N_{jt} claims, where N_{jt} is the number of properties damaged in zip code j and month t. For example, if there were 100 properties damaged in a zip code from a specific fire, we keep the 100 largest claims in that zip code in the month of the fire under the assumption that the largest fire claims result from properties damaged by wildfires.

For each claim i in this sample, we calculate underpayment and the resulting deficit as the following:

Underpaid_{*ijt*} = $\mathbb{1}\left(\operatorname{Claim}_{ijt} < \operatorname{Coverage} A \operatorname{Limit}_{ijt}\right)$ Deficit_{*ijt*} = Coverage A Limit_{*ijt*} - Claim_{*ijt*}.

Underpayment measures the share of policies that received fewer claims than their estimated replacement cost value, as measured by the Coverage A limit. The deficit measures the magnitude of deviation between claims and coverage. When aggregated across our sample of claims, which approximates the set of damaged properties facing a total loss, the deficit measures the extent of the shortfall between claims and replacement cost.

Restricting the sample of claims by the number of properties damaged conservatively measures underpayment and deficits. If a fire claim due to reasons other than a wildfire

²⁷This restriction matches our mortgage repayment analysis sample of single-family residences.

²⁸In total, we exclude 10 percent of policies. Specifically, we exclude guaranteed replacement cost coverage, which accounts for less than 4 percent of policies, and outliers in coverage limits and claim-to-limit ratios that account for the remainder of the exclusions.

exists in our sample, it replaces a claim resulting from a wildfire that is of a lower amount.²⁹ Therefore, any bias will overestimate the total claims paid resulting from the wildfire. Consequently, any observed underpayment in our sample will be a lower bound for the true underpayment rate. We also assume that the Coverage A limit is the actual replacement cost of a destroyed property. Studies have found the insurer-estimated RCV that dictates coverage limits is often much lower than the actual costs of rebuilding (Dixon et al., 2018; Klein, 2018, 2023). Therefore, our estimate of coverage deficits would be lower than the true deficit between the insurance settlement and the costs of rebuilding.

5.3 Insurance Underpayment and Deficits

We find a high prevalence of underpayment and associated deficits. Table 3 shows that between 2018 and 2020, approximately 40 percent of claims were underpaid. The resulting deficit is \$122,912, or 28 percent, of coverage limits. That is, the largest claims in zip codes where we observe fire damage received 72 cents per dollar of their respective Coverage A limits. Conditioning on underpaid claims, the deficit jumps to 83 percent. While these figures are high, they are similar to results from surveys of affected homeowners that show 42 percent to 66 percent of respondents would not be fully covered to replace or rebuild their home, with average deficits ranging from \$163,000 to \$375,000 (United Policyholders, 2022).

	All	2018	2019	2020
Deficit (share)	0.278	0.208	0.654	0.444
Deficit (\$)	122,912	$85,\!298$	411,990	244,812
Underpaid (share)	0.398	0.344	0.836	0.568
Deficit if Underpaid (share)	0.828	0.833	0.807	0.818
Deficit if Underpaid (\$)	$497,\!485$	482,344	524,734	$529,\!458$
Number of Claims	$11,\!430$	8,914	171	2,345
Number of Zip-Months	105	31	12	62

Table 3: Insurance Underpayment Prevalence and Deficit

Note: Table reports the underpayment rates and resulting deficits. A claim is considered underpaid if the settlement is below the Coverage A limit. The deficit is the difference between the Coverage A limit and claim amount. Deficits that are represented as shares are dollar weighted by the coverage limit. We include the largest N_{jt} claims in zip code j and month t, where N_{jt} is the number of observed wildfire-damaged properties, to identify the set of claims associated with wildfire damage.

²⁹Examples of fire claims due to reasons other than a specific wildfire include a kitchen fire occurring at the same time as a wildfire, fraudulent claims, or claims on undamaged properties requiring hazardous debris removal.

Next, we aggregate claims to the zip-code level and plot the correlation between average claim amounts and average Coverage A limits in Figure 5. Recall that California residents who intend to rebuild or purchase a new home are entitled to receive the lesser of the full coverage limit and quoted replacement costs under CA Insurance Code §2051.5, which would suggest claims at or close to \$1 per dollar of coverage (the 45-degree solid black line). The figure shows that almost all zip codes are, on average, underpaid. Only six zip codes fall at or above \$1 of claims paid per dollar of coverage. Accounting for extended RCV, only two zip codes' payments would exceed the coverage limit plus 20 percent additional coverage, as represented by the dotted line.



Figure 5: Zip-Code Level Coverage and Claims

Mean Coverage A Limit (\$ Thousands)

Note: The solid black 45-degree line represents the counterfactual, in which claim settlements equal the Coverage A limit. The dotted line represents the counterfactual of 20 percent extended replacement cost value coverage (ERCV), in which claims equal \$1.20 per dollar of Coverage A. Over 80 percent of policyholders hold at least 20 percent ERCV coverage. We include the largest N_{jt} claims in zip code j and month t, where N_{jt} is the number of observed wildfire damaged-properties.

We consider the following hypothesis that could lead to large observed underpayment rates and deficits. In an effort to reduce costs, insurers commonly send initially low settlement offers that cover only the ACV, deducting the depreciation of the house's materials from replacement costs. To calculate the ACV, insurers require several details about the structure, such as the quality of finishes and size of cabinets. When such details are not available, insurance adjusters rely on aggregated characteristics of the zip code housing the property. Households may accept these offers because of a lack of information or unwillingness to enter the required negotiation process to secure a higher settlement based on the property's individual damages. In the event of large fires that require quick processing of claims, it is likely that the dollar amount of claims would not vary greatly within zip codes. To examine this hypothesis, we estimate the following fixed-effects regression.

$$\operatorname{Claim}_{ijt} = \beta \operatorname{Coverage} A \operatorname{Limit}_{ijt} + \gamma X_{ijt} + \lambda_j + \lambda_t + \epsilon_{ijt}$$
(2)

In Equation 2, X_{ijt} includes additional extended replacement cost coverage and the policy premium rate per \$100,000 of coverage to account for policy quality. The fixed effects isolate the identifying variation to be as close to within-fire as our data allow by using zip code and month fixed effects. The β from this regression can be interpreted as the additional dollar of claims a household with an additional dollar of coverage receives for the same fire and same quality of policy. If claims reflect the rebuilding costs of the individual properties, as estimated by the Coverage A limits, we expect $\hat{\beta}$ to be close to 1. Moreover, the fixed-effects estimate $\hat{\beta}$ should not differ substantially from the ordinary least squares (OLS) estimate unless the property's zip code is a critical component of determining the share of the coverage limit households receive in claims.

However, both of these hypotheses are rejected. Table 4 shows that the fixed-effects estimate of the correlation between coverage limits and claims is weak, as $\hat{\beta}$ is statistically indifferent from zero. In addition, the correlation drops dramatically when introducing aggregated controls, such as extended RCV and policy price, and zip code fixed effects. These results support the hypothesis that the initial ACV settlement offer varies more by zip code characteristics than by the individual property replacement costs, as estimated by coverage limits.

5.4 Underinsurance and Rebuilding Costs

In addition to underpayment, households may be underinsured, meaning that the actual costs of rebuilding exceed the total coverage available through the policy. As a result, even if the settlement is not underpaid, households face a coverage gap and would receive insurance settlements that do not cover the cost of rebuilding (Dixon et al., 2018; United Policyholders, 2022). In such a case, the household faces strong incentives to prepay mortgages on damaged homes instead of rebuilding. While we cannot test for all causes of underinsurance, we empirically test whether higher-than-anticipated costs of rebuilding lead households to prepay

	OLS	OLS with Controls	Fixed Effects
Coverage A (\$)	0.3872*	0.1920*	0.0303
	(0.1537)	(0.0748)	(0.0550)
ERCV (Additional \$)		-0.0037	-0.0568
		(0.0658)	(0.0293)
Policy Premium		618.2^{***}	-64.07*
		(120.9)	(29.74)
Month-Year	No	No	Yes
Zip Code	No	No	Yes
Policy Type	No	No	Yes
Observations	11,430	11,430	11,430

Table 4: Correlation of Claims and Limits Within Zip Codes

Note: Table reports OLS and fixed-effects estimates of $\hat{\beta}$ from Equation 2. ERCV (Additional \$) represents the additional dollar amount extended replacement cost value coverage provides to the specific policy beyond the Coverage A limit. Policy premium measures the annual price per \$100,000 of Coverage A limit. The last column includes fixed effects for the month and year of loss, zip code, and policy type (homeowners or dwelling fire). We include the largest N_{jt} claims in zip code j and month t, where N_{jt} is the number of observed wildfire-damaged properties.

at higher rates. Since policy limits are fixed through the policy year, the coverage gap is increasing in reconstruction costs. Specifically, we use the triple difference design in Equation 3, where we interact property damage with proxies of high reconstruction costs, Z_{ift} .³⁰ A positive estimate of γ_{1t} , in which the prepayment rate is increasing in reconstruction costs, supports underinsurance as a determinant of our observed prepayment spikes.

$$Y_{ift} = \sum_{t=-24; t \neq -1}^{t=24} \beta_{1t} Damaged_{ift} + \sum_{t=-24; t \neq -1}^{t=24} \beta_{2t} Inside Fire Zone_{ift}$$
(3)
+
$$\sum_{t=-24; t \neq -1}^{t=24} \gamma_{1t} Damaged_{ift} \times Z_{ift} + \sum_{t=-24; t \neq -1}^{t=24} \gamma_{2t} Inside Fire Zone_{ift} \times Z_{ift}$$
+
$$\beta_3 X_{ift} + \lambda_{ft} + \lambda_{ft} \times Z_{ift} + \lambda_i + \epsilon_{ift}.$$

³⁰For continuous variables, such as log damaged properties, we standardize the values so that coefficients can be interpreted as the change in the dependent variable from a 1 standard deviation increase in the Z variable. We interact time fixed effects with our measures of heterogeneity to correct for estimation bias from multiple treatments, as discussed by de Chaisemartin and D'Haultfoeuille (2022).

First, a surge in demand for construction is common after large natural disasters. Homeowners in areas where the concentration of fire damage is greater should face a higher cost to rebuild. The cost to rebuild may exceed the policy limit by more than the allowed adjustments under extended RCV. Simultaneously, insurers may be financially stressed to pay out more insurance claims, providing further incentives to negotiate lower settlements with households. We estimate Equation 3 with the log number of damaged properties inside the fire perimeter as Z_{ift} . As shown in the top panel of Figure 6, estimates of $\hat{\gamma}_{1t}$ are positive and significant, confirming that the prepayment rates for damaged homes after a fire increase with fire severity. The role of frictions after fires does seem to be driven by the most severe fires. We see that the magnitude of the differential effect by fire severity is smaller but statistically significant when we exclude the Camp Fire from our estimation sample (Figure A.8).

Second, we estimate whether prepayment rates differ by the reconstruction costs based on the year the home was built. California substantially changed its wildfire standards in building codes during the 1990s and in 2008, such that these reforms require the use of fire resistant materials throughout the structure (e.g., roof, doors, windows, decks) and maintenance of the home's defensible space (e.g., vegetation around the home) that reduce the likelihood of damage given fire exposure (Baylis and Boomhower, 2021). As a result, the majority of new homes built after these regulations in California were required to comply with the stronger wildfire standards. Owners of older homes built before these regulations are required to comply with current wildfire standards during any substantial reconstruction, such as a roof replacement. Therefore, the cost to rebuild an older home will be substantially higher than the cost for a newer home, as the replacement costs will include additional costs to comply with new wildfire standards. We hypothesize that incentives to prepay are stronger for older homes than newer homes, as it is less likely that the insurance settlement will fully cover the higher cost of rebuilding. Estimates of $\hat{\gamma}_{1t}$ where Z_{ift} indicates whether the property was built before 1991 confirm this story. The bottom panel of Figure 6 shows that homes built before 1991, when stricter wildfire codes were first introduced, are more likely to prepay than homes built after 1991.

6 Welfare Implications

Our estimates of underpayment and deficits have strong welfare implications for households affected by a fire. We calculate a back-of-the-envelope change in household equity from before to after the fire based on received claims and measure any resulting shortfall relative to the amount households are entitled to under California Insurance Code §2051.5. The financial position of the household does not completely characterize household welfare, but a change





(a) Log of Properties Damaged \times Fire Damage



(b) Built pre-1991 \times Fire Damage

Note: Figure illustrates the coefficients for Z_{ift} interacted with damaged (γ_{1t}) and undamaged (γ_{2t}) parcels estimated from the event study specification in Equation 3. Both plots present the differential impact on prepayment after the fire for large fires as measured by number of properties damaged (top panel) and for loans on properties built before 1991 when stronger wildfire building codes were introduced (bottom panel). Coefficients estimate the marginal effect of a 1 standard deviation change in Z_{ift} . Regression controls for fire-month and loan fixed effects. Robust 95% confidence intervals are indicated by the shaded regions. in equity will underestimate total welfare losses under the assumption of a negative change in the nonmonetary components of welfare, given the stress associated with losing one's house to a fire and relocating.

Under Insurance Code §2051.5, if a household rebuilds or purchases a new home, California residents are entitled to the lesser of the full coverage limit and the quoted costs to rebuild the damaged home in the original location.³¹ The counterfactual change in equity from before to after the fire for households that receive the full payment under California law is therefore the following:³² If the household rebuilds, owners pay p_{ijt}^R in reconstruction costs. If the household opts to purchase a new home instead, the value of the existing property drops by the amount of damage incurred.

$$\Delta e_{ijt}^* = \min\{(1 + \text{ERCV}_{ijt}) \cdot \text{Coverage A Limit}_{ijt}, p_{ijt}^R\} \\ - \mathbb{1}(\text{Rebuild}_i) \cdot p_{ijt}^R - \mathbb{1}(\text{New Purchase}_i) \cdot \text{Damages}_{ijt}.$$

Unfortunately, we do not observe all the components to calculate the above. Instead, we estimate the shortfall between the actual change in equity, Δe_{ijt} , and counterfactual change in equity as

$$\Delta e_{ijt} - \Delta e_{ijt}^* = \text{Claim}_{ijt} - \max\{\text{Coverage A Limit}_{iit}, \text{Claim}_{ijt}\}.$$
(4)

Intuitively, this calculation assumes that, for claims that were not underpaid, households receive their entitled payout and thus do not face a shortfall. This calculation is likely underestimating the shortfall for households receiving claims exceeding limits, as we assume their settlement equals the full replacement cost. For underpaid claims, the shortfall between the actual and counterfactual equity change is simply the deficit.

We also estimate the actual change in equity assuming there is no rebuilding. This assumption is not as strong as it seems because very few households rebuild. Only 8 percent of the damaged properties file a construction permit in the two years following the fires between 2018 and 2020. Assuming that property damages lead to a total loss of structure value, we calculate the change in equity when a household does not rebuild as

$$\Delta e_{ijt}^{NR} = \text{Claims}_{ijt} - \text{Structure Value}_{ijt}.$$

³¹In the case of a new purchase, the amount entitled is also capped by the price of the new home, but we assume that a new home will be costlier than replacing the destroyed structure of the existing home.

 $^{^{32}}$ This calculation ignores deductibles, as they are very small compared with the total loss of a structure. The mean deductible is \$1,878, and the median is \$1,000.

We use zip-code level structure values, as estimated by Davis et al. (2021). The household gains the claim amount, which can be applied toward the mortgage. Note that we are assuming there is no additional loss of value to the land in the case of a fire. If the land value drops further, our calculated change in equity is an upper bound.

As shown in Table 5, households are financially worse off by \$199,290 to \$296,318 than if they received their entitled payout under California law, assuming caps on the payout are either Coverage A or ERCV. For reference, the actual change in equity assuming no rebuilding is a gain of \$21,998.³³ However, this improvement in the household's equity position does not imply that the household is better off. The household still has to purchase new housing and pay off any existing debt or tax obligations on the damaged house. While Insurance Code §2051.5 was designed to assist households with these financial stresses associated with disasters, the large shortfall implies a substantial welfare loss that households are legally protected against.

	N	Coverage A	ERCV	Δe_{ijt}
All	10,106	-199,290	-296,318	21,998
2018	$8,\!697$	-165,863	-251,867	48,602
2019	165	-433,958	-604,739	-209,706
2020	1,244	-401,858	-566,173	-133,264
Camp Fire Only	6,998	-73,841	-124,130	130,183
Excluding Camp Fire	$3,\!108$	-481,754	-684,018	-221,592

Table 5: Equity Shortfall Relative to Legal Entitlement

Note: Shortfall amounts as calculated from Equation 4. We show the calculations assuming that the legal entitlement under Insurance Code §2051.5 is capped by either the Coverage A limit or the ERCV. The change in equity in the last column assumes the household does not rebuild. Each row provides the same calculation for different subsamples of our claims data. We include the largest N_{jt} claims in zip code j and month t, where N_{jt} is the number of observed wildfire-damaged properties. We further restrict the claims to zip codes where an estimate of structure value is available from Davis et al. (2021).

A few factors determine these large shortfalls, although it is difficult to empirically identify their relative contributions. First, households and insurers are often uninformed of the protections under Insurance Code §2051.5. To meet the demand of claims processing

³³This equity gain is entirely driven by the Camp Fire, which is a unique case in which we observe at least 2,000 fewer claims than properties damaged. We assume this difference is due to homeowners not holding insurance (United Policyholders, 2022). If instead those properties did hold insurance and did not receive claim payments, the underpayment would imply a larger equity loss due to the Camp Fire than we calculate.

after large disasters, insurers rely on out-of-state adjusters who may not know the details of the state law protecting residents and instead dispute that a household would get the full coverage amount only if they rebuild. While other states have similar laws, such as valued policy laws, California is one of few states that both face large and frequent disasters and entitle residents to the full coverage amount even if they do not rebuild. Second, households may face credit constraints in purchasing a new home because their equity is illiquid in the form of burned land. As a result, lenders may be unwilling to provide credit to households for a seemingly second home. With limited access to credit, households may not be able to prove to the insurer that they will indeed purchase a new home.

Regardless of the reasons behind these shortfalls, households face large welfare losses relative to legal entitlements that have existed since 2004. Increasing insurer compliance, through information or enforcement, would largely improve households' financial condition after wildfires, which continue to increase in frequency and severity.

Lenders, Servicers, and Investors While we show large welfare losses for households, other agents in the mortgage face potential losses as well. For lenders, servicers, and investors in mortgage-backed securities, insurance effectively shifts the default risk associated with natural disasters toward prepayment risk. Even if damaged properties rarely default after a wildfire in our loan sample, the increase in prepayment can lead to losses. In the case of a prepayment, lenders lose anticipated interest revenue and servicers lose recurring servicing fees. Moreover, the temporary reduction in housing supply due to fire damage would impact the lender's ability to replace the mortgage in its portfolio. While investors in mortgage-backed securities are protected by the geographic diversification of their portfolios, a prepayment leads to a loss in the security's cash flow as the GSEs repay investors at par instead of the anticipated return.³⁴ Lenders and servicers also face operational costs to verify whether their collateral was indeed damaged.

However, total losses to lenders may be modest given the small number of mortgages affected by wildfires relative to the lender's portfolio. While we cannot explicitly calculate the total losses lenders, servicers, and investors face, we do test whether the marginal damaged home in a mortgage portfolio would increase the loss of interest revenue. We estimate heterogeneous effects on prepayment for loans with different interest rates (Equation 3). Banks would prefer borrowers with lower interest rates prepay to minimize lost interest revenue and to potentially add a new mortgage with a marginally higher interest rate. The estimate of $\hat{\gamma}_{1t}$ in the top panel of Figure 7 opposes bank preferences, as loans with higher

³⁴Furthermore, a share of the GSE losses get borne by private investors in credit risk transfer securities associated with the reference mortgage pool underlying the mortgage-backed securities.

interest rates have higher likelihoods of prepayment after a fire. Loans with a 1 standard deviation higher interest rate have a 3 percentage point higher likelihood of prepayment.

This pattern could still be beneficial to lenders, servicers, and investors if higher interest rates are correlated with borrower risk, which would result in a borrower pool with lower risk after high-interest mortgages prepay. The bottom panel of Figure 7 does not support this hypothesis and shows that there is no heterogeneous effect in prepayment across borrower risk, as measured by credit score. In fact, less risky borrowers with higher credit scores are slightly more likely to prepay, resulting in a borrower pool with higher risk.³⁵

These results suggest that prepayment after wildfires is not beneficial for lenders, servicers, and investors, as they face losses in interest revenue. Taken together with the implications for household welfare, wildfires and insurance market frictions that lead to prepayment suggest the potential for financial loss for all agents in mortgage markets.

7 Conclusion

We construct a novel database that merges property-level damage inspections from 79 California wildfires with mortgage performance and evaluates the impact of wildfires on mortgage repayment. This paper finds that 90-day delinquency rates and prepayment rates increase significantly for damaged properties soon after a fire. In contrast, repayment trends for undamaged properties, including those inside the fire's perimeter, remain unchanged. The timing and magnitude of the increase in home sales or refinances after a fire do not explain the immediate 16 percentage point increase in prepayment. We conclude that insurance payouts drive the large increase in prepayments for damaged properties after a wildfire, similar to the case of Hurricane Katrina (Gallagher and Hartley, 2017).

Our results highlight that wildfires present a larger risk to mortgage markets than the previous research finds. Measurement of fire damages at the property level shows that the direct effects of fire damage on repayment dwarf any spillover effects on undamaged homes inside the fire perimeter. As we estimate a reduced-form parameter that includes any protective effects of insurance and government aid, these risks for mortgages on damaged homes are potentially larger if safeguards weaken. These findings regarding the immediate aftermath of wildfires have implications for several stakeholders.

For households, the large increase in prepayment is an indicator for the lack of insurance coverage. We find substantial underpayment rates and deficits, in which the insurance settlement is lower than coverage limits, which are meant to proxy the costs of rebuilding.

³⁵We estimate the same heterogeneous effects for the delinquency outcome, as shown in Figure A.7. Borrowers with higher interest rates and lower credit scores are more likely to be past due by 90 days or more.

Figure 7: Differences in Prepayment by Interest Rate and Credit Score





Note: Figure illustrates the coefficients for Z_{ift} interacted with damaged (γ_{1t}) and undamaged (γ_{2t}) parcels estimated from the event study specification in Equation 3. Both plots present the differential impact on prepayment after the fire for loans with different interest rates (top panel) and borrowers with different credit scores (bottom panel). Coefficients estimate the marginal effect of a 1 standard deviation change in Z_{ift} . Regression controls for fire-month and loan fixed effects. Robust 95% confidence intervals are indicated by the shaded regions. Even if households receive the full coverage limit, demand surges drive up the reconstruction cost and lead to higher prepayment, as households find rebuilding costs to exceed those estimated by their existing coverage. Such insurance market frictions lead households to experience large financial shortfalls relative to their entitlements under California law.

For lenders, servicers, and investors, insurance effectively shifts the default risk associated with natural disasters toward prepayment risk. Even if damaged properties rarely default after a wildfire in our loan sample, the increase in prepayment can lead to lost interest revenue. In fact, we observe that, among damaged properties, mortgages with higher interest rates are more likely to be prepaid.

For policymakers, our findings imply that damage mitigation efforts before fires may effectively reduce the risks wildfires present to the mortgage market. Stronger wildfire codes for building standards largely reduce the likelihood of property destruction, and we estimate negative impacts on mortgage repayment only among damaged properties (Baylis and Boomhower, 2021).³⁶ Prepayment can also indicate a long-term reduction in housing supply if insurance payments are not used to restore damaged properties after a disaster as intended. Therefore, additional research is necessary to understand prepayment risks and the use of insurance funds after disasters.

Last, for researchers, this study emphasizes the need to obtain more precise measures of physical climate risks. As 56 percent of properties and 59 percent of mortgages within the fire perimeter in our sample remain undamaged, the use of larger geographic areas, such as fire perimeters, to proxy for property damage introduces substantial measurement error. Resulting estimates would be attenuated and may understate the true impact of fires on mortgage repayment. Our research motivates a greater focus on the local effects of natural disasters to fully understand the consequences of physical climate risks.

³⁶del Valle et al. (2022) show that damage mitigation efforts may also substitute for post-disaster spending, at least in the case of Hurricane Harvey.

References

- Addoum, J. M., Eichholtz, P. M. A., Steiner, E. and Yönder, E. (2021). Climate Change and Commercial Real Estate: Evidence from Hurricane Sandy. Working paper, SSRN, doi:10.2139/ssrn.3206257.
- An, X., Biswas, S., Hossain, M., Tarlin, S. and Zhang, C. (2022). A Bottom-Up Approach to Climate Stress Tests: The Effect of Hurricanes on Mortgage Repayment. Research brief, Federal Reserve Bank of Philadelphia.
- An, X., Gabriel, S. A. and Tzur-Ilan, N. (2023). The Effects of Extreme Wildfire and Smoke Events on Household Financial Outcomes. Working paper, SSRN, doi:10.2139/ssrn.4353113.
- Atreya, A. and Czajkowski, J. (2019). Graduated Flood Risks and Property Prices in Galveston County. *Real Estate Economics* 47: 807–844, doi:10.1111/1540-6229.12163.
- Atreya, A., Ferreira, S. and Kriesel, W. (2013). Forgetting the Flood? An Analysis of the Flood Risk Discount over Time. Land Economics 89: 577–596, doi:10.3368/le.89.4.577.
- Bach, A. and Wade, D. (2018). What They Don't Know Can Help You: California Policyholder Protections Insurers and Adjusters May "Overlook". *Forum* 48: 24–26.
- Baker, A. C., Larcker, D. F. and Wang, C. C. Y. (2022). How Much Should We Trust Staggered Difference-in-Differences Estimates? *Journal of Financial Economics* 144: 370– 395, doi:10.1016/j.jfineco.2022.01.004.
- Bakkensen, L. and Barrage, L. (2022). Going Underwater? Flood Risk Belief Heterogeneity and Coastal Home Price Dynamics. *Review of Financial Studies* 35: 3666–3709, doi:10.1093/rfs/hhab122.
- Bakkensen, L., Phan, T. and Wong, R. (2023). Leveraging the Disagreement on Climate Change: Theory and Evidence. Working Paper 23-01, Federal Reserve Bank of Richmond, doi:10.21144/wp23-01.
- Baldauf, M., Garlappi, L. and Yannelis, C. (2020). Does Climate Change Affect Real Estate Prices? Only If You Believe In It. *The Review of Financial Studies* 33: 1256–1295, doi:10.1093/rfs/hhz073.
- Battiston, S., Mandel, A., Monasterolo, I., Schütze, F. and Visentin, G. (2017). A Climate Stress-Test of the Financial System. *Nature Climate Change* 7: 283–288, doi:10.1038/nclimate3255.

- Baylis, P. W. and Boomhower, J. (2021). Mandated vs. Voluntary Adaptation to Natural Disasters: The Case of U.S. Wildfires. Working Paper 29621, National Bureau of Economic Research, doi:10.3386/w29621.
- Bernstein, A., Gustafson, M. T. and Lewis, R. (2019). Disaster on the Horizon: The Price Effect of Sea Level Rise. *Journal of Financial Economics* 134: 253–272, doi:10.1016/j.jfineco.2019.03.013.
- Billings, S. B., Gallagher, E. A. and Ricketts, L. (2022). Let the Rich be Flooded: The Distribution of Financial Aid and Distress After Hurricane Harvey. *Journal of Financial Economics* 146: 797–819, doi:10.1016/j.jfineco.2021.11.006.
- Boomhower, J., Fowlie, M., Gellman, J. and Plantinga, A. (2023). How Are Insurance Markets Adapting to Climate Change? Risk Selection and Regulation in the Market for Homeowners Insurance. Working paper.
- Borusyak, K., Jaravel, X. and Spiess, J. (2021). Revisiting Event Study Designs: Robust and Efficient Estimation. Working paper, arXiv, doi:10.48550/ARXIV.2108.12419.
- Boustan, L. P., Kahn, M. E., Rhode, P. W. and Yanguas, M. L. (2020). The Effect of Natural Disasters on Economic Activity in U.S. Counties: A Century of Data. *Journal of Urban Economics* 118: 103257, doi:10.1016/j.jue.2020.103257.
- Bradt, J. T. and Aldy, J. E. (2022). Private Benefits from Public Investment in Climate Adaptation and Resilience. Working paper.
- Caetano, C. and Callaway, B. (2023). Difference-in-Differences with Time-Varying Covariates in the Parallel Trends Assumption. Working paper, arXiv, doi:10.48550/ARXIV.2202.02903.
- California Department of Insurance (2021). Residential Insurance: Homeowners and Renters. http://www.insurance.ca.gov/01-consumers/105-type/95-guides/03-res/res-ins-guide.cfm, Last accessed on 2022-10-26.
- Callaway, B. and Sant'Anna, P. H. C. (2021). Difference-in-Differences with Multiple Time Periods. *Journal of Econometrics* 225: 200–230, doi:10.1016/j.jeconom.2020.12.001.
- Canals-Cerdá, J. and Roman, R. (2021). Climate Change and Consumer Finance? A Very Brief Literature Review. Discussion Paper, Supervisory Research Forum Spotlight, Federal Reserve Bank of Philadelphia.

- Cengiz, D., Dube, A., Lindner, A. and Zipperer, B. (2019). The Effect of Minimum Wages on Low-Wage Jobs. *Quarterly Journal of Economics* 134: 1405–1454, doi:10.1093/qje/qjz014.
- Clerc, L., Bontemps-Chanel, A. L., Diot, S., Overton, G., Albergaria, S. Soares de, Vernet, L. and Louardi, M. (2021). A First Assessment of Financial Risks Stemming from Climate Change: The Main Results of the 2020 Climate Pilot Exercise. Tech. rep., Autorité de Contrôle Prudentiel et de Résolution, Banque de France.
- Cookson, J. A., Gallagher, E. and Mulder, P. (2023). Money to Burn: Wildfire Insurance via Social Networks. Working paper, SSRN, doi:10.2139/ssrn.4535190.
- Davis, M. A., Larson, W. D., Oliner, S. D. and Shui, J. (2021). The Price of Residential Land for Counties, ZIP Codes, and Census Tracts in the United States. *Journal of Monetary Economics* 118: 413–431, doi:https://doi.org/10.1016/j.jmoneco.2020.12.005.
- de Chaisemartin, C. and D'Haultfoeuille, X. (2022). Two-way Fixed Effects and Differencesin-Differences Estimators with Several Treatments. Working Paper 30564, National Bureau of Economic Research, doi:10.3386/w30564.
- del Valle, A., Scharlemann, T. C. and Shore, S. H. (2022). Household Financial Decision-Making After Natural Disasters: Evidence from Hurricane Harvey. Working Paper 2022-015, Federal Reserve Board of Governors Finance and Economics Discussion Series, doi:10.17016/FEDS.2022.015.
- Deryugina, T., Kawano, L. and Levitt, S. (2018). The Economic Impact of Hurricane Katrina on Its Victims: Evidence from Individual Tax Returns. American Economic Journal: Applied Economics 10: 202–233, doi:10.1257/app.20160307.
- Dixon, L., Tsang, F. and Fitts, G. (2018). The Impact of Changing Wildfire Risk on California's Residential Insurance Market. Report, California Natural Resources Agency.
- Du, D. and Zhao, X. (2020). Hurricanes and Residential Mortgage Loan Performance. Working Paper 2020-04, Office of the Comptroller of the Currency.
- Fang, L., Li, L. and Yavas, A. (2023). The Impact of Distant Hurricane on Local Housing Markets. Journal of Real Estate Finance and Economics 66: 327–372, doi:10.1007/s11146-021-09843-3.
- Farrell, D. and Greig, F. (2018). Weathering the Storm: The Financial Impacts of Hurricanes Harvey and Irma on One Million Households. Report, JPMorgan Chase Institute.

- Fried, S. (2022). Seawalls and Stilts: A Quantitative Macro Study of Climate Adaptation. The Review of Economic Studies 89: 3303–3344, doi:10.1093/restud/rdab099.
- Gallagher, J. and Hartley, D. (2017). Household Finance After a Natural Disaster: The Case of Hurricane Katrina. American Economic Journal: Economic Policy 9: 199–228, doi:10.1257/pol.20140273.
- Gallagher, J., Hartley, D. and Rohlin, S. (2023). Weathering an Unexpected Financial Shock: The Role of Federal Disaster Assistance on Household Finance and Business Survival. Journal of the Association of Environmental and Resource Economists 10: 525– 567, doi:10.1086/721654.
- Gibson, M. and Mullins, J. T. (2020). Climate Risk and Beliefs in New York Floodplains. Journal of the Association of Environmental and Resource Economists 7: 1069–1111, doi:10.1086/710240.
- Goodman-Bacon, A. (2021). Difference-in-Differences with Variation in Treatment Timing. Journal of Econometrics 225: 254–277, doi:10.1016/j.jeconom.2021.03.014.
- Grippa, P. and Mann, S. (2020). Climate-Related Stress Testing: Transition Risks in Norway. Working Paper 2020/232, International Monetary Fund, doi:10.5089/9781513559674.001.
- Groen, J. A., Kutzbach, M. J. and Polivka, A. E. (2020). Storms and Jobs: The Effect of Hurricanes on Individuals' Employment and Earnings over the Long Term. *Journal of Labor Economics* 38: 653–685, doi:10.1086/706055.
- Hino, M. and Burke, M. (2021). The Effect of Information about Climate Risk on Property Values. Proceedings of the National Academy of Sciences 118: e2003374118, doi:10.1073/pnas.2003374118.
- Issler, P., Stanton, R., Vergara-Alert, C. and Wallace, N. (2021). Housing and Mortgage Markets with Climate-Change Risk: Evidence from Wildfires in California. Working paper.
- Johnson, H. (2022). California's High Housing Costs Have Created a Million "House Rich" Millionaires. Blog post, Public Policy Institute of California.
- Kaufman, L. (2021). Millions in Fire-Ravaged California at Risk of Losing Home Insurance. Los Angeles Times, September 26, 2021.
- Keys, B. J. and Mulder, P. (2020). Neglected No More: Housing Markets, Mortgage Lending, and Sea Level Rise. Working Paper 27930, National Bureau of Economic Research, doi:10.3386/w27930.

- Klein, K. S. (2018). Minding the Protection Gap: Resolving Unintended, Pervasive, Profound Homeowner UUnderinsurance. Connecticut Insurance Law Journal 25: 34–116.
- Klein, K. S. (2023). The Unnatural Disaster of Insurance, Underinsurance, and Natural Disasters. *Connecticut Insurance Law Journal*, Forthcoming.
- Kousky, C., Palim, M. and Pan, Y. (2020). Flood Damage and Mortgage Credit Risk: A Case Study of Hurricane Harvey. *Journal of Housing Research* 29: S86–S120, doi:10.1080/10527001.2020.1840131.
- Lougee, B. J., Morley, T., Watson, M. et al. (2018). The Road to Cyberinfrastructure at the Federal Reserve Bank of Kansas City. CADRE Technical Briefings doi:https://doi.org/10.18651/tb/tb1802.
- McConnell, K., Whitaker, S. D., Fussell, E., DeWaard, J., Curtis, K., Price, K., St. Denis, L. and Balch, J. (2021). Effects of Wildfire Destruction on Migration, Consumer Credit, and Financial Distress. Working paper, Federal Reserve Bank of Cleveland, doi:10.26509/frbcwp-202129.
- Murfin, J. and Spiegel, M. (2020). Is the Risk of Sea Level Rise Capitalized in Residential Real Estate? *Review of Financial Studies* 33: 1217–1255, doi:10.1093/rfs/hhz134.
- NOAA (2022). U.S. Billion-Dollar Weather and Climate Disasters. Database, National Centers for Environmental Information, National Oceanic and Atmospheric Administration.
- Ortega, F. and Taṣpinar, S. (2018). Rising Sea Levels and Sinking Property Values: Hurricane Sandy and New York's Housing Market. *Journal of Urban Economics* 106: 81–100, doi:10.1016/j.jue.2018.06.005.
- Panjwani, A. (2022). Underwater: The Effect of Federal Policies on Households' Exposure to Climate Change Risk. Working paper.
- Ratcliffe, C., Congdon, W., Teles, D., Stanczyk, A. and Martín, C. (2020). From Bad to Worse: Natural Disasters and Financial Health. *Journal of Housing Research* 29: S25–S53, doi:10.1080/10527001.2020.1838172.
- Reinders, H. J., Schoenmaker, D. and van Dijk, M. A. (2020). A Finance Approach to Climate Stress Testing. Working paper, SSRN, doi:10.2139/ssrn.3573107.
- Roncoroni, A., Battiston, S., Escobar-Farfán, L. O. L. and Martinez-Jaramillo, S. (2021). Climate Risk and Financial Stability in the Network of Banks and Investment Funds. *Journal of Financial Stability* 54: 100870, doi:10.1016/j.jfs.2021.100870.

- Roth, J., Sant'Anna, P. H., Bilinski, A. and Poe, J. (2023). What's Trending in Difference-in-Differences? A Synthesis of the Recent Econometrics Literature. *Journal of Econometrics* 235: 2218–2244, doi:https://doi.org/10.1016/j.jeconom.2023.03.008.
- Sever, C. and Perez-Archila, M. (2021). Climate-Related Stress Testing: Transition Risk in Colombia. Working Paper 2021/261, International Monetary Fund, doi:10.5089/9781513599205.001.
- Sun, L. and Abraham, S. (2021). Estimating Dynamic Treatment Effects in Event Studies with Heterogeneous Treatment Effects. *Journal of Econometrics* 225: 175–199, doi:10.1016/j.jeconom.2020.09.006.
- United Policyholders (2022). Roadmap to Recovery Surveys. https://uphelp.org/media/ surveys/, Last accessed on 2022-10-26.
- Vermeulen, R., Schets, E., Lohuis, M., Kölbl, B., Jansen, D.-J. and Heeringa, W. (2021). The Heat Is On: A Framework for Measuring Financial Stress Under Disruptive Energy Transition Scenarios. *Ecological Economics* 190: 107205, doi:10.1016/j.ecolecon.2021.107205.

A Appendix

All results presented in the figures and tables of this Appendix are derived from calculations based on the data described in Section 2 of the paper unless otherwise noted.

Nist National Institute of Standards and Technology U.S. Department of Commerce						
1 COLLECTION D	ETAILS					
Incident Name:	Camp Fire	Recording Da	ite: 12, 2	18 Time	Recorded:	1:55a 497 2110
0	Photo Release I	Form Approved:	□ Yes □ No	ZIN/A Pho	to Numbers: 1	
2 ADDRESS	Fixed DRV	/Travel Trailer	🗆 Manufa	ctured Home		
2 C						
Street Number	Street Name			Unit No.		
Paradise	CA		(an)			
City	State	ZIP			-	
Property Owner Last Nam	ne (if known)	_				
3 DAMAGE TO ST	RUCTURE				-	
Extent of Damage	: D Affected	Minor	Major	Destro	ved ⊡t	No Damage
Ignition/Damage	Embers	Radiation / Conv	vection	Undetermined		
Damaged Feature	Accoments					
Feature Damaged Feature	amaged	Feature	Damaged	F	eature	Damaged
Roof		Eaves	25	v	findows	M
Roof Valley / Transitions		Gutters		D	oors	
Dormers		Siding/Walls	2	D	ecking	
Window Detaile	Single Pane	Frame	Damage	X Vinyl	□ Wood	Metal
Nindow Details.	Serve toth panes	M Seal D	amage	L Fiberglass	L) Other	
Deers Datallas	U Window Damage	Frame	Damage	12 Vinyl	□ Wood	Metal
Door Details:	Door Damage	Seal D	amage anars	Fiberglass	C Other	D N/D
Decking Details:	Top Side X Pos	sts 🗆 Bottom Si	de	□ Wood	Composite	Other
4 <u>NOTES/DESCR</u> (brief description of d - defended - many was - many d - shew burned	ted undurs, so freed inthes with behind	n, details for Section me with b m all s	s) sta pano suer y ho	rs me		
have m \$	sides burned out	• 18				
- Moosel MA		1 A A A A A A A A A A A A A A A A A A A	· · · · ·			

Figure A.1: Example Damage Report for Camp Fire

Source: NIST Investigation of the California Camp Fire, Fire Research Division, National Institute of Standards and Technology.

Category of Damage	Definition	Examples
Affected (1-9%)	Minimal damage to the exterior and/or contents of the building. Building is habitable/usable and requires mostly cosmetic repairs.	Partially damaged shingles or siding, but roof structure is intact. Cosmetic damages such as paint discoloration, blistering or melted siding. Broken windows. Gutter damage. Damage to an attached structure like a deck, porch, carport, or patio cover.
Minor (10-25%)	Encompasses a wide range of damage that does not affect the structural integrity of the building. Building is not habitable/usable.	Nonstructural damage to roof components (e.g. roof covering, fascia board, soffit, flashing, and skylight). Nonstructural damage to the interior wall components to (e.g. drywall and insulation). Nonstructural damage to exterior components (e.g. door and windows. Substantial damage to exterior covering (e.g. siding, vinyl or stucco). Damage to mechanical components (e.g. furnace, boiler, water heater, HVAC, etc.).
Major (26-50%)	A building that has sustained significant structural damage and requires extensive repairs. Building is not habitable/usable.	Failure or partial failure of structural elements to include rafters, ceiling joists, ridge boards, etc. Failure or partial failure to structural elements of the walls to include framing, sheathing, etc.
Destroyed (>50%)	The building is a total loss, or damaged to such an extent that repair is not feasible.	Complete failure to major components (foundation, walls, roof, etc.). Two or more walls destroyed and roof substantially damaged. Only the foundation remains. The building will have to be torn down and rebuilt as it is unsafe.

Figure A.2: Classification of Damage Categories

Source: Camp Incident Damage Inspection Report CABTU 016737



Figure A.3: Distribution of Damage Severity

Note: Figure plots the distribution of damage categories reported in the DINS database.



Figure A.4: Post-Wildfire Loan Status Flows: Three and Six Months After Fire

(a) Loan Status Three Months After Fire



(b) Loan Status Six Months After Fire

Note: Figure illustrates the decomposition of loans within wildfire boundaries based on whether or not the home was damaged by the wildfire. The top panel shows mortgage status three months post-fire (current, delinquent, foreclosed, or prepaid) while the bottom panel shows mortgage status six months post-fire.



Figure A.5: Robustness to Omitting Camp Fire

Note: Figure illustrates the coefficients for damaged (β_{1t}) and undamaged (β_{2t}) parcels estimated from the event study specification in Equation 1 after excluding the Camp Fire in 2018. The top panel presents results for delinquency and the bottom panel presents results for prepayment. Regression controls for fire-month and loan fixed effects. Robust 95% confidence intervals are indicated by the shaded regions.

Figure A.6: Robustness to Choice of Control Group: Properties 5 to 6 Miles Outside Fire Perimeter



Note: Figure illustrates the coefficients for damaged (β_{1t}) and undamaged (β_{2t}) parcels estimated from the event study specification in Equation 1 using homes in the 5- to 6-mile ring outside the fire perimeter as the control group. The top panel presents results for delinquency and the bottom panel presents results for prepayment. Regression controls for fire-month and loan fixed effects. Robust 95% confidence intervals are indicated by the shaded regions.

Figure A.7: Differences in Delinquency by Interest Rate and Credit Score





Note: Figure illustrates the coefficients for Z_{ift} interacted with damaged (γ_{1t}) and undamaged (γ_{2t}) parcels estimated from the event study specification in Equation 3. Both plots present the differential impact on delinquency after the fire for loans with different interest rates (top panel) and borrowers with different credit scores (bottom panel). Coefficients estimate the marginal effect of a 1 standard deviation change in Z_{ift} . Regression controls for fire-month and loan fixed effects. Robust 95% confidence intervals are indicated by the shaded regions.



Figure A.8: Log Damages Heterogeneity: Robustness to Omitting Camp Fire

Note: Figure illustrates the coefficients for Z_{ift} interacted with damaged (γ_{1t}) and undamaged (γ_{2t}) parcels estimated from the event study specification in Equation 3. The plot presents the differential impact on prepayment after the fire for large fires as measured by number of properties damaged after excluding the Camp Fire in 2018. Coefficients estimate the marginal effect of a 1 standard deviation change in Z_{ift} . Regression controls for fire-month and loan fixed effects. Robust 95% confidence intervals are indicated by the shaded regions.